Author's response to reviews

Title: Effectiveness of an Ultrasound Training Module for Internal Medicine Residents

Authors:

Mira T Keddis (keddis.mira@mayo.edu)
Michael W Cullen (cullen.michael@mayo.edu)
Darcy A Reed (reed.darcy@mayo.edu)
Andrew J Halvorsen (halvorsen.andrew@mayo.edu)
Furman S. McDonald (mcdonald.furman@mayo.edu)
Paul Y. Takahashi (takahashi.paul@mayo.edu)
Anjali Bhagra (bhagra.anjali@mayo.edu)

Version: 2 Date: 29 March 2011

Author's response to reviews: see over
March 29, 2011

BMC Medical Education

Dear Sir/Madam,

Please find enclosed a submission to the following category: Research. The manuscript is entitled “Effectiveness of an Ultrasound Training Module for Internal Medicine Residents.” The results presented in this paper have not been published previously in whole or part, except in abstract form. All authors have read and approved submission of the manuscript.

Corresponding author takes full responsibility for the manuscript. Neither of the authors has any relevant conflicts of interests with regards to this manuscript. Please let me know if you have any questions or concerns regarding this submission.

Sincerely,

Anjali Bhagra M.D.
Department of Primary Internal Medicine
Mayo Clinic
200 First Street SW
Rochester, MN
55905
Tel: 507-284-5187
Email: bhagra.anjali@mayo.edu